**Reviewer’s code:** 06110739  
**SPECIFIC COMMENTS TO AUTHORS**

Surgical resection of brain glioma is mostly used therapy for the treatment of brain glioma. However, damage to functional areas of brain may usually occur due to the special location of brain glioma leading to high incidence of complications and even death. Unfortunately, few studies discussed the usage of comprehensive nursing in patients with brain glioma. In this study, the authors explored the efficacy of comprehensive nursing in patients who underwent operation for brain glioma and analyze its effects on the incidence of postoperative complications, psychological state and quality of life in patients with brain glioma. The study is very well written. The results are interesting and well discussed. A minor editing is required before final publication of the study. Thank you.

Answer: Thank you for your comments. The authors edited the manuscript carefully.

---

**Reviewer’s code:** 06099468  
**SPECIFIC COMMENTS TO AUTHORS**

This is an interesting study about the effects of comprehensive nursing on postoperative complications, mental status and quality of life in patients with glioma. The manuscript is well written, and study is well designed. The references should be edited and updated.

Answer: Thank you for your comments. The references were edited and updated.